

The current issue and full text archive of this journal is available at www.emeraldinsight.com/2044-1398.htm

Effects of differences of opinions and short-sale constraints on the dual listed Chinese shares

Zhenmin Fang and Xin Jiang Department of Economics and Finance, City University of Hong Kong, Kowloon, Hong Kong

Abstract

Purpose – The purpose of this paper is to study the effects of short-sale constraints and differences of opinions on the price premium of dual listed Chinese A-H shares.

Design/methodology/approach – The analysis mainly follows the Miller's model, which indicates that the relaxation of stringent short-sale constraint could reduce the upward bias in stock prices. Following the literature, the paper uses the idiosyncratic return volatility and monthly turnover rate as two main proxies of differences of opinions.

Findings – This study shows that the high level of A-share differences of opinions will lead to the high price premium of A-share portfolio with the short-sale constraint in the A-share market. However, the high level of H-share differences of opinions has no effect on the price premium of H-share portfolio and has also positively contributed to the A-share price premium. The price premium of shorted A-share portfolio is declined more significantly than those of non-shorted ones after the relaxation of short-sale constraint in the A-share market.

Research limitations/implications – The findings in this study provide further evidence that dual listed Chinese A-shares with high level of differences of opinions and short-sale constraints tend to be overvalued.

Practical implications – This study supports Miller's hypothesis that with the control of short-sale constraint, the high level of differences of opinions could lead to the high degree of overvaluation of A-share portfolio. The market capitalization and book-to-market ratio of A-shares also generate significant positive effect to the A-share price premium. Finally, the introduction of short-sale mechanism in A-share market could partially eliminate the mispricing of dual-listed A-shares and improve the price efficiency of A-share market.

Originality/value – This study is mainly focused on the joint effects of differences of opinions and short-sale constraints on the A-share price premium. The new short-sale policy in A-share market in March 2010 provides us an opportunity to study the effect of relaxation of stringent short-sale constraint on the A-share price premium. In the literatures so far, all studies assumed A-shares are strictly prohibited to be sold short.

Keywords Differences of opinions, Short-sale constraint, A-share price premium, Dual listed shares, China, Shares

Paper type Research paper

1. Introduction

The benefits of international diversification attract free capital to move across borders. Such benefits prompt investors to pay higher prices for foreign stocks than they would

JEL classification - G12, G14, G15

The authors thank the journal Editor for his invitation and encouragement. The authors also thank the reviewers for their careful reading and constructive comments and suggestions.



China Finance Review International Vol. 3 No. 1, 2013 pp. 61-89 © Emerald Group Publishing Limited 2044-1398 DOI 10.1108/20441391311290785

Dual listed Chinese shares

61

pay for the domestic stocks. However, Chinese dual-listed shares have long been considered as a puzzle that the prices of A-shares which are exclusively held by Chinese domestic investors are usually higher than that of their corresponding foreign shares. The anti-intuitive phenomenon was first documented by Bailey and Jagtiani (1994). Since then, the early literatures were focused on explaining why there are price dispersions between Chinese A- and B-shares. In the last ten years, the academics shifted their attention to focus on the price dispersions between dual-listed A- and H-shares. H-shares are denominated in Hong Kong dollars issued, supervised and traded on the Stock Exchange of Hong Kong. The first H-share, Tsingtao Brewery, was listed in Hong Kong on July 15, 1993. Subsequent to its listing of H-shares in Hong Kong, the corresponding A-share was listed on the Shanghai Stock Exchange (SSE) on August 27, 1993. Since the establishment of dual-listed A- and H- share markets, it is found that the trading prices of H-shares are persistently lower than that of their corresponding A-shares. Why are there persistent A-share price premiums over that of H-shares? What are the main factors contributed to the direction and magnitude of the price dispersions between A- and H-shares? It is our interest in this study to tackle these questions.

Short-sale constraint has been considered as the main factor for the A-share price premiums in the Chinese stock markets. According to Miller's (1977) well-cited study, given the short-sale constraints, negative information cannot be immediately incorporated into stock prices. As a consequence of price discovery delay, stock prices may only reflect the opinions of the optimists, leading to the stock overvaluation. In addition to the short-sale constraint, another cornerstone of Miller's overvaluation hypothesis is the difference of opinion. For any given degree of short-sale constraint, the more heterogeneous the expectations, the greater the price and return bias. Likewise, given the amount of divergence in expectations, the higher the constraint on short sales, the greater the price and return bias. The imperfect disclosure of public information and the speculative nature of Chinese individual investors are the main factors causing high heterogeneous opinions in the Chinese stock market.

Before March 2010, short sale was not allowed for all stocks on Chinese stock exchanges. The A-share prices would mainly reflect the expectations of the most optimistic investors and thus generate high price premiums. On March 20, 2010, the China Securities Regulatory Commissions (CSRC) relaxed the stringent short-sale constraint in A-share market, allowing Chinese investors to sell short the composite stocks of the SSE 50 Index and the Shenzhen Stock Exchange (SZSE) 40 Index via designated security brokers. Such a relaxation caused the pessimists' opinions to be incorporated into the A-share prices. As a result, the price discovery efficiency of A-share market is improved.

In the literature, there are few studies focused on the joint effects of short-sale constraint and differences of opinions on the price premiums between dual-listed A- and H-shares. The aim of this study is to:

- provide evidence that the A-share price premiums could be explained by the joint effects of short-sale constraint and differences of opinions; and
- demonstrate that the introduction of short-sale mechanism in A-share market would narrow the price dispersions between dual-listed A- and H-shares and improve the price discovery.



CFRI

3.1

62

In our study, we first examine the effects of differences of opinions on A-H share price dispersions with strict short-sale constraint before the introduction of short-sale policy in A-share market. Then we use event study to examine the effect of the relaxation of short-sale constraint on the A-share price premium. Our results reveal that before the introduction of short-sale mechanism in A-share market, the differences of opinions in both A- and H-share markets have significant positive relationships with the A-share price premiums. The relationships hold even when we control for the market capitalization, book-to-market ratio, past momentum returns and other potential factors. The results of event study show that the price premiums of shorted A-shares are declined more than that of non-shorted ones after the new short-sale policy taking place, indicating that the introduction of short-sale policy improves the price discovery of A-share market.

The rest of the paper is organized as follows. Section 2 provides the literature reviews. Section 3 describes the data, variables and testing hypotheses. Section 4 provides empirical tests for the impacts of differences of opinions and short-sale constraint on the A-share price premiums. Section 5 provides the summary and conclusions.

2. Literature reviews

Our research is mainly based on two streams in the literatures. One is the study on the possible explanations for the A-share price premiums and the other one is the study on the joint effects of differences of opinions and short-sale constraint on the capital market performance.

2.1 Studies on A-share price premiums

Regarding the A-share price premiums, most of the literatures are focused on the A-share premiums over B- and H-shares. Several competent hypotheses have been used to explain the A-share premiums. Sun and Tong (2000) showed that the A-share premiums could be explained by economic principles and macroeconomic factors. They found that H-shares and red-chip stocks could be viewed as the substitutions of B-shares and the supply of H-shares and red-chip stocks were positively related to the A-B share price dispersions. In addition, the sensitivity of foreign investors to the inflation rate and change in the official reserves also contributed to the A-share price premiums. Karolyi and Li (2002) found that the fundamental attributes of stocks matter for the magnitude of dual-listed A-B share price dispersions. Specifically, the decline in B-share discounts was concentrated in small capitalization stocks and those with substantial past return momentum around the regulatory changes in 2001. It was not related to the firm's risk and liquidity attributes.

Recently, academics tried to explain the price dispersions of A-shares from the perspective of market microstructure and information environment. Chan *et al.* (2008) constructed three measures of information asymmetry based on the market microstructure models and found that those measures could explain 45 percent of the variation of A-B share premiums. By isolating the impacts of disclosure practices on informational risk, Tang (2011) found that disparity in the public disclosure of information to domestic and foreign investors created a meaningful difference in investors' average information precision across A- and B- shares which contributed to the cross-sectional variation in price differences.

Besides, other factors such as investor sentiment and changes of exchange rate expectation have also been proved to be attributable to the long-term A-share premiums.



Wang and Jiang (2004) found that A-share returns are subjected to the market-specific risk and investor sentiment that are specific to the Chinese stock market, while H-share returns are subjected to the risk and investor sentiment in both Hong Kong and China stock markets. Using intraday high-frequency data, Chong and Su (2006) investigated the co-movement of A- and H-shares in a short period. Their results suggested that the stock markets of China and Hong Kong are segmented and for the small portion of stocks with co-movement, the A-share market played a major role in the price discovery process. By studying the price behavior of 13 dual-listed A-H shares, Li *et al.* (2006) found that the premiums on A-shares relative to H-shares are highly related to the contemporaneous premium of A-share local market relative to H-share local market, especially during the period of Asian financial crisis. Other factors such as the spread of savings rates between Hong Kong and China also contributed to the A-H share price differentials.

Arquette *et al.* (2008) found that the discounts attached to Chinese securities, whether trading as ADRs or H-shares appeared to have been significantly influenced by changes in both exchange rate expectation and investor sentiment during 1998-2006. Their conclusions were generally maintained under additional allowances for market capitalization, dividend payment and the company fixed effects in panel regressions. Burdekin and Redfern (2009) studied the role of investor sentiment in A-share market and provided the evidence that investor sentiment exerted consistently significant effects on the discounts attached to all Chinese foreign shares, including B-shares, H-shares and ADRs and the effects remained very robust even after controlling for other explanatory factors such as expected exchange rate movement and liquidity level. Callen *et al.* (2009) investigated the price disparity between A-shares and B- and H-shares. They decomposed the unexpected price disparity into differences in expected return news and cash flow news and found that the difference of expected return news dominated that of cash flow news in driving the variation of price disparity, suggesting that market news rather than firm specific news caused the price disparity of dual-listed shares.

Cai *et al.* (2011) developed a non-linear Markov error correction model to capture several important dimensions for the dual-listed A-H share dynamic pricing. With respect to the long-run A-share premium, their results revealed that the higher the A-share premium is, the higher the relative difference will be in the market's information asymmetry and opinion divergence. They also proved that the expectation of RMB revaluation also contributed to the long-run A-share premium. In addition, they noted that the recent financial crisis gave rise to a short-term reversal towards a narrowing of the A-share premium. The recent paper of Tong and Yu (2012) offered corporate governance explanation to the A-share premium. Their results suggested that A-share premium is relative high for firms with weak corporate governance characters.

2.2 Studies on joint effects of differences of opinions and short-sale constraints

Miller (1977) established the theoretical foundation for the empirical research in this area. In his study, he asserted that uncertainty and risk implied divergence of opinions. In a market with little or no short selling, the demand for a particular security would come from the minority who hold the most optimistic expectations about it. Since divergence of opinion was likely to increase with risk, it is quite possible that expected returns would be lower for risky securities, rather than higher, which was predicted by the traditional asset pricing models.



CFRI

3.1

Till now, most of the applications of Miller's overvaluation model are concentrated in US markets. Though academics adopted different proxies for differences of opinions and short-sale constraints, their conclusion are qualitatively similar. For example, Chen et al. (2002) adopted the breadth of mutual fund ownership as a measure of the level of short-sale constraint and found that decreases (increases) in breadth of ownership led to low (high) subsequent returns. Diether et al. (2002) took the dispersion in analysts' forecast as the proxy and showed that stocks with higher dispersion in analysts' earnings forecast earned lower future returns than otherwise similar stocks. This effect was most pronounced in small stocks and stocks that had performed poorly over the past year. Johnson (2004) showed that relationships between forecast dispersion and subsequent returns documented by Diether et al. (2002) could be explained by the financial leverage. Recent studies examined the joint effects of short-sale constraint and differences of opinions on the stock performance and the conclusions are more consistent with Miller's hypothesis. Boehme et al. (2006) showed evidence that the significant overvaluation for stocks were subject to both differences of opinions and short-sale constraint. Stocks were not systematically overvalued when either one condition was not met.

Nagel (2005) adopted the institutional ownership as the proxy for the short-sale constraint. He found that holding size fixed, the underperformance of stocks with high market-to-book ratio, analyst forecast dispersion, turnover or volatility was most pronounced among stocks with low institutional ownership. Berkman *et al.* (2009) also adopted the institutional ownership as the indirect measure of short-sale constraint and other five distinct proxies for the differences of opinions (turnover rate, return volatility, forecast dispersion, list age and income volatility) and found that stock returns with high differences of opinions are more negative within the subsample of stocks that are most difficult to sell short.

For the applications of Miller's overvaluation hypothesis in Chinese stock market, Mei *et al.* (2003) analyzed the joint effects of short-sales constraint and heterogeneous beliefs on dual-listed Chinese A-B share prices and trading volumes. They found that the A-B share price difference was positively related to the domestic turnover rate. Both price difference and turnover rate increased with the idiosyncratic return volatility and decreased with the float of domestic shares. Their results indicated that investors' speculative motives could help to explain a significant fraction of the price difference. Chan *et al.* (2010) investigated the effect of short-sale eligibility of H-shares on A-H share premiums. They found that when the market went down, the prices of shorted H-shares decreased more than those of non-shorted ones, resulting in larger A-H share premiums. As the short-sale eligibility allowed more investors to trade, the trading volumes were high for shorted H-shares which contributed to the higher A-H price premiums.

Compared with the study of Chan *et al.* (2010), our study is focused on the joint effects of differences of opinions and short-sale constraint on the A-share price premiums. Besides, the relaxation of the stringent short-sale constraint in A-share market in March 2010 provided us an opportunity to further study the effects of short-sale constraint on the A-share price premiums. While in the literatures so far, all studies assumed A-shares are strictly prohibited to be sold short.

3. Data, variables and testing hypotheses

In this section, we describe the data and variable constructions and propose the testing hypotheses.



3.1 Data and variable constructions

CFRI

3.1

66

Our data includes price premiums of firms that are dual-listed on the SSE and the Stock Exchange of Hong Kong from January 2001 to March 2011. At the end of March 2011, there are totally 55 pairs of A-H dual-listed shares in our study[1]. The A-share transactions, returns and firm-characteristics data are retrieved from China Security Market Research Database (CSMAR) and the H-share data are retrieved from the Datastream.

We define A-share price premium *PREM* as:

$$PREM_{i,t} = \frac{P_{i,t}^{A} - P_{i,t}^{H}}{P_{i,t}^{A}}$$
(1)

where P^A is the A-share monthly closing price, P^H is its corresponding H-share monthly closing price which has been converted to RMB by monthly exchange rate, and *i* identifies the firm and *t* identifies the current calendar month. We exclude the observations if either monthly closing price is missing. The main explanatory variables in our study are the proxies for the differences of opinions in both Shanghai and Hong Kong stock markets. We adopt two variables as the main proxies of differences of opinions according to the literature and data availability.

The first proxy is the idiosyncratic monthly return volatility (*SIGMA*) which is computed as the standard deviation of daily idiosyncratic return. We exclude the firm-month observation if it has less than five daily returns in one specific month. Following Mei *et al.* (2003), we measured the idiosyncratic return $\mu_{i,t}$ as the residual return in the following CAPM model:

$$R_{i,t} = \beta_0 + \beta_1 M K T_{i,t} + \beta_2 M K T_{i,t-1} + \mu_{i,t}$$
(2)

where R is the daily A-share (H-share) return, MKT is the daily Shanghai Composite Index (Hang Seng China Enterprise Index) return and i identifies the firm and t identifies the current calendar month.

The second proxy is the monthly turnover rate (TURN) obtained by:

$$TURN_{i,t} = \frac{\sum_{j=1}^{n} DST_j}{MSO}$$
(3)

where DST is the number of daily shares traded and MSO is the monthly shares outstanding and n is the number of trading days in the current calendar month. For A-shares, we use number of tradable shares instead of total shares outstanding as the denominator to obtain the monthly turnover rate. Again we exclude the firm-month observation if there are less than five daily trading records in a specific month.

In the robustness test, we use one alternative proxy *RETVOL* to examine the effects of differences of opinions on the A-H price premiums. *RETVOL* is the monthly return volatility which is measured by the standard deviation of a firm's daily stock excess return in a month in which the excess return is measured as the difference between A-share (H-share) daily return and the corresponding return of daily Shanghai Composite Index (Hang Seng China Enterprise Index). In our analysis, we also use several variables to control for the potential factors influencing the long-term A-share premiums. We obtain the stock market capitalization as the product of monthly closing



price and the number of total shares outstanding in June of each year. In our study, we take the natural logarithm of the market capitalization to avoid the potential skewness. We also control for the book-to-market factor to ensure that the effect of differences of opinions is distinct from the glamour anomaly. Book-to-market ratio is obtained by dividing the book value of the firm to its market value of equity at the end of previous fiscal year. To capture the effects of past return momentum, we calculate cumulative buy-and-hold return for previous one month (*RET1*), six months (*RET26*) and penultimate six months (*RET712*), respectively, for each firm.

Following Fama and French (1992), we estimate the stock beta as the sum of slopes in the regression of following CAPM model:

$$R_{i,t} = \beta_0 + \beta_1 M K T_{i,t} + \beta_2 M K T_{i,t-1} + \mu_{i,t}$$
(4)

where R is the monthly A-share (H-share) raw return, MKT is the monthly Shanghai Composite Index (Hang Seng China Enterprise Index) return and i identifies the firm and t identifies the current calendar month. In order to obtain reliable beta estimation, there must be at least 12 of the 36 monthly returns in firm-month observations preceding to the estimation month. From the estimation of above equation, we denote:

Beta =
$$\beta_1 + \beta_2$$
.

Many of the existing literatures have pointed out the importance of market condition and exchange rate change in realigning A- and H-share prices (Arquette *et al.*, 2008; Cai *et al.*, 2011). In order to control for these two factors and also to rule out the possibility that our differences of opinions proxies reflect the rapidly changing fundamentals or changing perception of these fundamentals, we calculate another two sets of control variables, *MKTRET* and *EXP*. Specifically, *MKTRET*^A and *MKTRET*^H are the monthly returns of Shanghai Composite Index and Hang Seng China Enterprise Index (market returns in A- and H-share markets), respectively, and *EXP* is measured as:

$$EXP_t = \frac{NDF_t - SP_t}{SP_t} \tag{5}$$

where *NDF* is the 12-month non-deliverable forward price of RMB against US dollar and *SP* is the spot exchange rate between RMB and US dollar.

In the event study, the dummy variable, *SHORT*, is equal to one if the A-share observation is allowed to be shorted after March 2010, otherwise it is zero. Besides eliminating the A-share observations with insufficient data available, some variables with extreme values will largely affect the empirical results. Thus, we further winsorize *PREM, SIGMA, TURN, DISP, RETVOL, MV, BM, RET1, RET26, RET712* and *Beta* at 1 and 99 percent levels.

3.2 Testing hypotheses

The relatively simple environment in Chinese stock market (strictly enforced restriction on short-sales, segmentation of A- and H-share markets and the lack of derivative instrument) before the new short-sale policy in March 2010 facilitates us to isolate the effect of short-sale constraint. We expect that the high heterogeneous opinions of A-share prices will lead to the high degree of overvaluation of A-shares during the sample period from January 2001 to February 2010. Thus, our first hypothesis is:

CFRI 3,1

68

H1. Before the introduction of short-sale mechanism in A-share market, the high level of A-share differences of opinions will lead to the high level of A-share price premium.

However, the short-sale regulations vary widely across A- and H-share markets. The latter put shorted securities in an official list which is revised over time. In addition, the Hong Kong market is well established, more open and rigorous in terms of listing requirements and information disclosure than that of the China stock market. The different institutional settings regarding the short-sale policies make the impacts of heterogeneous opinions in Hong Kong quite different from that in Shanghai. With heterogeneous beliefs and no short-sale constraints in the H-share market[2], both optimistic and pessimistic investors determine the equilibrium H-share prices. Thus, in the H-share market, we expect that the differences of opinions may not lead H-share prices to deviate from their fundamental values and cause the overvaluation of H-shares. However, with stringent short-sale constraint in the A-share market, pessimistic investors are unable to sell short the overvalued A-shares and the equilibrium A-share prices will have a positive bias as we stated in the H1. As a result, our second hypothesis is:

H2. Since H-shares are not subjected to the strict short-sale constraint, the high level of differences of opinions in H-share market may not lead to the high H-share price relative to the corresponding A-share price and thus also contribute positively to the A-share price premium.

The policy launched in March 2010 allowed short sales for some A-shares. The introduction of short-sale policy in A-share market made the price of shorted A-shares to reflect not only the opinions of optimistic investors but also the opinions of pessimistic investors. As a result, we expect that the shorted A-shares become less overvalued. Since there are no fundamental changes in the H-share market during the event period, we conjecture that A-share premium would decline sharply for shorted A-shares. And for the non-shorted A-shares, the price premiums would also be narrowed because of the improvement of overall information environment in the A-share market. However, the premium decline of non-shorted A-shares should be less than that of shorted ones. Thus, the new policy would enhance the price discovery of shorted A-shares. Hence, our third hypothesis is:

H3. The price dispersions between A- and H-shares are narrowed more for shorted A-shares than those of non-shorted ones after the new short-sale policy is introduced in the A-shares market.

4. Empirical studies on A-H share price premiums

4.1 Descriptive statistics

Table I presents the summary statistics of price premium variables (Panel A) and their correlation matrix (Panel B) in A- and H-share markets. It is observed that there are very large A-share price premiums, *PREM*, during the sample period with mean 45.85 percent, median 50.35 percent and standard deviation of 28.63 percent. Regarding the proxies of difference of opinion, it seems there is no big difference between idiosyncratic return volatility (*SIGMA*) in A-share and H-share markets. The mean (median) of *SIGMA_a* and *SIGMA_h* is 1.76 percent (1.58 percent) and 2.02 percent (1.83 percent), respectively. However, the monthly turnover rate (*TURN*) is quite



	RET712_a	0.0027 0.0489 0.0489 0.0348 -0.1045 0.0109 0.1237 1.0000	$\begin{array}{c} RE7712_h\\ -0.1093\\ 0.0769\\ 0.1559\\ 0.0376\\ -0.2613\\ 0.0543\\ 0.0543\\ 0.1191\\ 1.0000\end{array}$	listed A-shares; ratio, past one- $RET26_h$ and timate past six-	Dual listed Chinese shares
	RET26_a	$\begin{array}{c} 0.0259\\ 0.1279\\ 0.0837\\ 0.0210\\ -0.0040\\ 0.0129\\ 1.0000\end{array}$	$\begin{array}{c} RET26_h\\ -\ 0.0778\\ 0.0778\\ 0.0166\\ 0.0168\\ 0.0168\\ -\ 0.0924\\ 0.0295\\ 1.0000 \end{array}$	rice premium of dual- ation, book-to-market $-I_h$, BM_h , $RET1_h$, onth return and penul	69
	RET1_a	$\begin{array}{c} 0.0382\\ 0.1535\\ 0.1535\\ 0.1429\\ -0.0041\\ 1.0000\end{array}$	$\begin{array}{c} RET1.h\\ -0.0450\\ 0.1127\\ 0.1127\\ 0.0119\\ -0.0017\\ 1.0000\end{array}$	urkets; <i>PREM</i> is the p rate, market capitaliz MA_h, TURN_h, M1 Ath return, past six-m	
	SD 5D 0.2863 0.0118 0.0118 0.3610 0.0118 0.3610 0.0118 0.3610 0.173 0.1731 0.1373 0.1469 0.1644 0.1373 0.1469 0.1644 0.1064 0.1373 0.14695 0.14695 0.16640000000000000000000000000000000000	-0.2024 -0.1043 0.0410 -0.0535 1.0000	BM_h 0.4423 0.4423 0.1985 - 0.0528 - 0.2580 1.0000	A- and H-share me monthly turnover narket, while <i>SIG</i> ratio, past one-moi potheses	
	Q3 0.6913 0.0525 0.0225 0.0225 0.0225 0.0225 0.0225 0.0225 0.0225 0.0325 0.0329 0.0329 0.0329 0.0329 0.0329 0.0329 0.0328 0.1947 0.2285 0.1947 0.2285 0.1947 0.2285 0.1947 0.2285 0.1947 0.2285 0.1947 0.2285 0.1947 0.2285 0.1947 0.2285 0.1947 0.0228 0.0229 0.0225 0.0226 0.0225 0.0225 0.0225 0.0229 0.0229 0.0229 0.0229 0.0229 0.0229 0.0229 0.0229 0.0229 0.0229 0.0229 0.0229 0.0228 0.0208 0.0008 00008 0008 00008 00008 000080000000	-0.2845 -0.2502 -0.1782 1.0000	MV_h - 0.3805 - 0.3405 - 0.1327 1.0000	rrelation matrix in c return volatility, y, in the A-share r m, book-to-market or our first two hy	
	tics Q_I Q_I 0.2596 0.2596 0.0107 0.0107 0.0107 0.0107 0.0125 0.0107 0.01273 0.0877 6.7371 1.7129 0.0877 6.7371 1.7729 0.0684 -0.0684 -0.0684 -0.0684 -0.1779 0.01779 -0.1779 -0.1779 -0.1779 -0.1779 -0.1779 -0.1033 -0.1779 -0.1779 -0.1779 -0.1033 -0.12282 -0.12282 -0.12282 -0.12282 -0.12282 -0.12282 -0.12282 -0.12282 -0.12282 -0.12282 -0.12282 -0.12779 -0.12282 -0.12779 -0.12282 -0.12779 -0.12282 -0.12779 -0.12282 -0.12779 -0.12779 -0.12779 -0.12779 -0.12779 -0.12779 -0.12779 -0.12782 -0.12779 -0.12779 -0.12782 -0.12779 -0.12779 -0.12782 -0.12779 -0.12782 -0.12779 -0.12779 -0.12779 -0.12779 -0.12779 -0.12779 -0.12782 -0.12779 -0.12782 -0.12782 -0.12779 -0.12782 -0.1278	0.2435 0.4380 1.0000	<i>TURN_h</i> 0.1480 0.3464 1.0000	riables and their co a is the idiosyncrati return, respectivel market capitalizati e rough evidence fi	
	anel A: summary statist Median 0.5035 0.5035 0.0158 0.0158 0.0158 0.0157 0.0157 0.0574 0.5247 0.6250 0.0070 0.0070 0.0072 0.00721 0.00721 0.00721 0.00721 0.00721 0.00653 SJGM_a	0.3878 1.0000	SIGMA_h 0.3410 1.0000	ics of price premium va <i>ET26_a</i> and <i>RET712_1</i> Itimate past six-month monthly turnover rate, tet; Table I provides th	
	P: Mean 0.1585 0.0176 0.0176 0.02062 0.3842 0.20622 0.3842 0.20622 0.3751 0.6727 0.6727 0.6727 0.6727 0.0109 0.0751 0.0109 0.0523 0.1759 PREM	1.0000	PREM 1.0000	the summary statistic BM_a , $RET1_a$, R h return and penul to return volatility, ic return volatility, in the H-share mark	
	Variable PREM SIGMA_a SIGMA_a SIGMA_h TURN_a TURN_a TURN_b MV_a (in RMB millions) BM_a BM_a BM_h RET1_a RET1_a RET1_a RET26_a RET26_a RET26_a RET26_a RET712_b RET712_b	PREM SIGMA_a TURN_a MV_a BM_a RET1_a RET712_a RET712_a	PREM PREM TURN_h MV_h BM_h BM_h RET1_h RET21_h RET712_h RET712_h	Notes: This table presents the $SGCMA_a$, $TURN_a$, MV_a , month return, past six-month $RET772_h$ is the idiosyncrat month return, respectively, in	Table I. Summary statistics of price premium variables and their correlation matrix in A- and H-share markets
ئىارات	المنسا رة للاست				www.

different in A- and H-share markets. The mean (median) of $TURN_a$ is 38.42 percent (26.46 percent), which is nearly two times as that of $TURN_h$ with mean (median) of 20.60 percent (15.71 percent). The high A-share turnover rate in our study is consistent with findings in most of the literatures.

Figure 1 shows the relationship between dual-listed A-share price premium and idiosyncratic monthly return volatility (*SIGMA*) in both A- and H-share markets from January 2001 to March 2011. And Figure 2 shows the relationship between dual-listed A-share price premium and monthly turnover rate (*TURN*) in both A- and H-share markets from January 2003 to March 2011. As shown in two figures, the A-share price premiums tend to have a downward trend. For example, Figure 1 shows that the premium has a very high value of 89.58 percent at the beginning of January 2001 and decreases to 21.85 percent in September 2010. However, the two proxies of differences of opinions, *SIGMA* and *TURN*, tend to be very volatile without a clear pattern in the period from January 2001 to March 2011.

From Table I, we also find that the market capitalization (MV) of A-share is much larger than that of the corresponding H-share. The mean (median) of MV_a is 113.79 million (20.49 million) which is much larger than MV_h with mean (median) of 44.50 million (6.06 million). The result of MV is not surprising as the trading of dual-listed A-H shares is more active in Shanghai than that in the Hong Kong stock market. As the result of the market capitalization difference, the dual-listed A-H shares have a relative high BM ratio in Hong Kong with mean (median) of 77.51 percent (62.50 percent), compared with 67.27 percent (52.47 percent) in Shanghai. In addition,



Figure 1. Dual-listed A-share price premium and idiosyncratic return volatility in A- and H-share markets

CFRI

3.1

70

Figure 2.

Dual-listed A-share price premium and monthly turnover rate in A- and H-share markets the past stock performance of H-share is usually better than that of A-share. Specifically, the average past one-month return RET1 is 1.09 percent for A-shares and 2.25 percent for H-shares. The average past six-month return RET26 is 5.91 percent for A-shares and 13.78 percent for H-shares. When we extend the return horizon to the penultimate six-month, the average return RET712 rises to 6.23 percent for A-shares and 17.59 percent for H-shares.

The Pearson correlation matrix of Table I provides the rough evidence to our first two hypotheses. It is observed that the levels of differences of opinions in both A- and H-share markets have positive relationships with A-share premiums. For example, the correlation coefficient between the price premium, PREM and SIGMA_a (SIGMA_h) is 0.3878 (0.3410). It indicates that the high idiosyncratic return volatility in both A- and H-share markets would lead to the high A-share price premiums. Similarly, the high correlation between *PREM* and monthly turnover rate *TURN* in both markets would have the similar effects on the price differentials of A-H shares. Considering the relationship between *PREM* and market capitalization *MV*, we find the correlation coefficient is -0.2845 in A-share market and -0.3805 in H-share market, indicating that the price premiums are negatively related to the market capitalization in both A- and H-share markets. By contrast, the growth opportunity and past stock performance in Shanghai and Hong Kong markets have generated opposite effects on A-share price premiums. Specifically, the A-share (H-share) BM ratio is negatively (positively) related to A-share price premiums. The positive correlations between PREM and RETs in A-share market indicate that if the dual-listed stocks performed well in A-share market in the past, the high A-share premiums would be expected. The negative correlations between *PREM* and *RETs* in H-share market show just the opposite.

4.2 H1 and H2 testing with portfolio analysis

In this section, we test the validity of H1 and H2 regarding the relationships between the level of differences of opinions and the magnitude of A-share price premiums. We first examine the relationships between the mean price premiums of A-share portfolios and the two main proxies of differences of opinions. The sample period for *SIGMA* and *TURN* is from January 2001 to March 2010. The portfolio analysis helps us to determine the qualitative effects of differences of opinions on the A-H share price dispersions in the market with a strict short-sale constraint. We then examine the effects with more precise Fama and MacBeth (1973) regression analysis in which we control for the pronounced size, book-to-market ratio and return momentum effect.

To study the effects of the relative magnitude of differences of opinions r_SIGMA and r_TURN on the A-share price premiums, we define:

$$r_SIGMA_{i,t} = \frac{SIGMA_{i,t}^A - SIGMA_{i,t}^H}{SIGMA_{i,t}^A}$$
(6)

$$r_TURN_{i,t} = \frac{TURN_{i,t}^A - TURN_{i,t}^H}{TURN_{i,t}^A}$$
(7)

where $SIGMA^A$ and $SIGMA^H$ are the idiosyncratic return volatility and $TURN^A$ and $TURN^H$ are the monthly turnover rate in A- and H-share markets, respectively, and *i* identifies the firm and *t* identifies the current calendar month. We divide firm-month



observations into two portfolios based on the absolute magnitude of *SIGMA* and *TURN* in both A- and H-share markets, respectively, and also based on the relative magnitude of r_SIGMA and r_TURN . For each month, we sorted our observations by the differences of opinion proxies. If the difference of opinion proxy (*SIGMA*, *TURN*, r_SIGMA or r_TURN) of the firm-month observation is smaller (larger) than that of the monthly median, then the firm will be falling into "low" ("high") sub-portfolio. For each "low" and "high" sub-portfolio, the mean A-share price premium is calculated for every calendar month. Finally, we compare the mean A-share premium between "high" and "low" sub-portfolios using "high-low". Table II reports the mean price premiums of A-share portfolios formed by means of two main proxies of differences of opinions.

The results based on the absolute magnitude of differences of opinions *SIGMA* and *TURN* in Table II verify both of our *H1* and *H2*. That is, we do observe positive relationships between the level of differences of opinions in both A- and H-share markets and the A-share price premiums. In fact, the A-share price premium increases monotonically with the level of differences of opinions in both A- and H-share markets. In the A-share market, for portfolios categorized by *SIGMA*, the A-share price premium increases from 47 percent with low level of *SIGMA* to 60.88 percent with high *SIGMA* value. For portfolios categorized by *TURN*, the A-share price premium increases from 41.63 percent with low level of *TURN* to 50.49 percent with high *TURN* value.

To test whether the relationships between mean A-share price premiums and differences of opinions are statistically significant, we calculate the differences of mean A-share price premiums between portfolios with high and low levels of differences of opinions. The results in Table II show that all differences are statistically significant at 1 percent level using a two-tailed *t*-test. In the A-share market, the high-low difference is 13.88 and 8.86 percent, respectively, for *SIGMA* and *TURN*. They are 11.83 and 7.07 percent, respectively, for *SIGMA* and *TURN* in

Portfolio	SIGMA_a	Mean price SIGMA_h	e premium of A <i>TURN_a</i>	-share portfolio <i>TURN_h</i>	r_SIGMA	r_TURN
Low	0.4700 ***	0.4804 ***	0.4163***	0.4254 ***	0.5441 ***	0.4567***
High	(23.34)	(23.97)	(26.63)	(26.61)	(32.56)	(29.04)
	0.6088 ***	0.5988 ***	0.5049***	0.4961 ***	0.5353 ***	0.4729***
High-low	(41.11)	(39.61)	(39.87)	(36.41)	(29.55)	(36.14)
	0.1388 ^{***}	0.1183 ^{***}	0.0886 ^{***}	0.0707***	- 0.0087	0.0162
	(14.22)	(12.01)	(8.97)	(5.27)	(- 0.35)	(0.79)

Notes: Significant at: *10, **5 and ***1 percent levels; this table reports the mean price premium of A-share portfolio by means of two proxies of differences of opinions; in each calendar year-month, firms with dual-listed A- and H-shares are divided into two portfolios based on the high or low level of *SIGMA* and *TURN* in both A- and H-share markets and also based on the high or low level of *r_SIGMA* and *r_TURN*, where *r_SIGMA* is the relative idiosyncratic return volatility and *r_TURN* is the relative monthly turnover rate in A-H share markets; we first calculate the average price premium of each portfolio based on *SIGMA* and *TURN*, respectively, in both A- and H-share markets and also based on *r_SIGMA* and *r_TURN*; difference in mean price premium of A-share portfolio based on *SIGMA* and *TURN*; difference in mean price premium of A-share portfolio (high-low in the table) based on *SIGMA* and *TURN* is both A- and H-share markets and based on *r_SIGMA* and *r_TURN* is evaluated with a two-sided *t*-test; White's (1980) heteroskedasticity-consistent *t*-statistics are shown in parentheses

Table II.

Portfolio analysis of mean price premium of A-shares based on proxy of differences of opinions

للاستشارات

www.man

 $\mathbf{72}$

the H-share market. The results demonstrate that the mean A-share price premiums of portfolios with high level of differences of opinions are significantly higher than that with low level of differences of opinions.

However, when we calculate the differences of mean A-share price premiums between portfolios with high and low levels of relative magnitude of differences of opinions, the differences are not statistically significant. As can be seen from Table II, the "high-low" difference is -0.87 percent for r_SIGMA and 1.62 percent for r_TURN and both differences are not statistically significant. The result is not surprising because the level of differences of opinions in both A- and H-share markets is positively related to the A-share premium. The high level of A-share differences of opinions drags down the H-share prices. When we adopt r_SIGMA and r_TURN in the analysis, the effects of A- and H-share differences of opinions offset each other so the A-share premium differences of opinions and the share premium differences of opinions offset each other so the A-share premium differences are not significant.

To further examine the effects of differences of opinions on the A-share price premiums, we adopt an independent two-dimensional categorized approach to take both A- and H-share differences of opinions into consideration simultaneously. Specifically, for each calendar year-month, we categorize A- and H-share *SIGMA* and *TURN* into two independent portfolios each. Then we merge these portfolios together based on *SIGMA* and *TURN*, respectively. This approach will leads to four double-categorized portfolios, respectively, based on *SIGMA* or *TURN*. Table II provides the mean price premiums of above double-categorized A-share portfolios.

Consistent with our H1 and H2 and the results in Table II, the results in Table III indicate that the differences of opinions in both A- and H-share markets are positively related to the A-share price premiums. The relationships hold even when we control for the effects of differences of opinions in either A- or H-share market. For example, in Panel A of Table III, when the A-share *SIGMA* is low, the difference of A-share price premiums between portfolios with high and low H-share *SIGMA* is 8.51 percent and significant at 1 percent level. On the other hand, when the H-share *SIGMA* is low, the difference of A-share price premiums between portfolios with high and low A-share *SIGMA* is 12.24 percent and also significant at 1 percent level. For the *TURN* in Panel B of Table III, when the A-share *TURN* is high, the difference of A-share price premiums between portfolios with high and low H-share *TURN* is 5.29 percent and significant at 5 percent level. On the other hand, when the H-share *TURN* is high, the difference of A-share price premiums between portfolios with high and low A-share the other hand, when the H-share *TURN* is high, the difference of A-share price premiums between portfolios with high and low A-share *TURN* is 7.34 percent and also significant at 5 percent level.

4.3 H1 and H2 testing with regression analysis

In this section, we examine the effects of differences of opinions on A-share price premiums in a regression framework after controlling for the market capitalization, book-to-market ratio and past return momentums in both A- and H-share markets:

$$PREM_{i,t} = \beta_0 + \beta_1 DIFF_{i,t}^A + \beta_2 DIFF_{i,t}^H + \beta_3 MV_{i,t}^A + \beta_4 MV_{i,t}^H + \beta_5 BM_{i,t}^A + \beta_6 BM_{i,t}^H + \beta_7 RET1_{i,t}^A + \beta_8 RET1_{i,t}^H + \beta_9 RET26_{i,t}^A + \beta_{10} RET26_{i,t}^H$$
(8)
+ $\beta_{11} RET712_{i,t}^A + \beta_{12} RET712_{i,t}^H + \mu_{i,t}$

74

		Low	High	High-lo
Panel A: mean	price premium of A	-share portfolio		
		ale ale	SIGMA_a	
SIGMA_h	Low	0.4391 ***	0.5581 * * *	0.1224
		(19.93)	(32.31)	(10.57)
	High	0.5208 ***	0.6357 * * *	0.1125
		(27.86)	(39.61)	(9.79)
	High-low	0.0851 ***	0.0753 * *	
		(7.34)	(6.91)	
Panel B: mean	price premium of A-	share portfolio		
			TURN_a	
TURN_h	Low	0.3916***	0.4688***	0.0772
		(23.77)	(21.08)	(4.09)
	High	0.4484 ***	0.5218***	0.0734
	0	(22.74)	(43.57)	(7.68)
	High-low	0.0568***	0.0529**	(
	8	(3.52)	(2.52)	

Table III.

Multivariate analysis of A-share price premium based on proxy of differences of opinions A-share portfolio based on an independent two-dimensional categorized approach to take both A- and H-share differences of opinions into consideration simultaneously; for each calendar year-month, we categorize A- and H-share idiosyncratic return volatility (*SIGMA*) and monthly turnover rate (*TURN*) into two independent portfolios each; we then merge these portfolios based on *SIGMA* and *TURN*, respectively; we first calculate the average price premium of each portfolio in every calendar-month categorized by *SIGMA* and *TURN*; we then calculate and report the mean price premium of A-share portfolio based on *SIGMA* (Panel A) and *TURN* (Panel B), respectively; the difference in mean price premium of A-share portfolio is evaluated with a two-sided *t*-test; White's (1980) heteroskedasticity-consistent *t*-statistics are shown in parentheses

where *PREM* is the mean A-share price premium, *DIFF*^A and *DIFF*^H are the specified proxies of differences of opinions (*SIGMA* and *TURN*) in A- and H-share markets, respectively, MV^A and MV^H are the market capitalizations and BM^A and BM^H are the book-to-market ratios in A- and H-share markets, respectively, $RET1^A$, $RET26^A$, $RET712^A$ and $RET1^H$, $RET26^H$ and $RET712^H$ are the past one-month return, past six-month return and penultimate past six-month return in A- and H-share markets, respectively, again *i* identifies firm and *t* identifies the calendar month. To be consistent with our portfolio analysis, the estimation period based on *SIGMA* and *TURN* are from January 2001 to March 2010. Table IV reports the estimation results of nine regression model specifications. For the convenience of interpretation, all variables are converted into percentages before regression.

According to our *H1* and *H2*, the predicted sign of regression coefficients of proxies of differences of opinions in both A- and H-share markets should be positive. The estimation results based on *SIGMA* and *TURN* in the first two columns of Table IV are consistent with our predictions. The *SIGMA* seems to have a much greater explanatory power for the dual-listed A-share price premiums compared with that of *TURN*. Both *SIGMA*^A and *SIGMA*^H in Model 1 could explain 20.84 percent of the variability of A-share price premiums.

Based on the estimation results from regression Models 3-7 in Table IV, we could infer that the stock market capitalization (MV), book-to-market ratio (BM) and past



تشارات		(VIRAD) F	A AGT TTY A	c	-	ı	c	1		
		1 (SIGMA)	2 (TURN)	cc	4	2	9	7	8 (SIGMA)	9 (TURN)
للاس	$DIFF_a$	8.87*** (1 / 1 2)	0.21^{***}						5.29 *** (6.66)	0.18***
	$DIFF_h$	5.26***	0.12^{**}						1.10^{*}	0.28***
à	MV_{-a}	(8.34)	(2.51)	9.04^{***}					(1.65) 12.44 ***	(8.23) 12.49 ^{***}
J	$h_{V}h_{V}$			$(16.53) - 15.36^{***}$					$(10.61) - 17.25^{***}$	(10.69) - 18.53 ***
	BM_{-a}			(-26.55)	-27.54***				$(-11.76) - 10.48^{***}$	(-12.24) -14.96^{***}
i	BM_h				(-12.93) 28.58 [*] **				(-2.93) 0.84	(-4.20) 2.19
	$RET1_a$				(21.48)	0.18**			(0.35) 0.17 ***	$(0.88) \\ 0.18^{***}$
	$RET1_h$					(2.21) - 0.22 ***			(2.77) - 0.16 ***	$(2.68) - 0.14^{**}$
	$RET26_a$					(-3.93)	0.04		(-2.86) 0.12 ***	(-2.27) 0.15 0.15
	$RET26_h$						$(1.03) - 0.13^{***}$		$(3.43) - 0.06^{**}$	$(4.88) - 0.12^{***}$
	$RET712_a$						(-4.40)	0.04	(-2.10) 0.06	(-4.14) 0.05
	$RET712_h$							$(0.98) - 0.11 ^{***}$	(1.89) - 0.02	$(1.67) - 0.04^{*}$
	Adjusted R^2 (%)	20.84	9.42	45.47	30.57	10.24	12.19	(-4.06) 10.71	(-1.01) 58.30	(-1.69) 62.48
	Notes: Significan opinions after cont variable in all nine TURV in all mode (MV), book-to-mar table shows the r (in parenthesis), w	t at: $*10$, $*5$ au trolling for the . . model specific als are specified list ratio (BM) , regression estii there the weigh	nd ***1 percent market capitaliz ations is the me by <i>DIFF</i> (<i>DIFF</i> part one-month mate of weight the correspond t	: levels; this tabl attion, book-to-r an price premui- Δ and $DIFF_{A}$ a return ($RETI$) ed Fama and Γ o the number o	le shows the regunation of A-shares, in A-shares, in A-and H-s, i, past six-mon Macbeth coeffi	gression results d past return m PREM in equa hare markets; t th return (RET cients and thei in each month	of A-share prid comentums in b tion (8); two pri- tion (8); two pri- tion (8); two pri- tion (8); two pri- 26) and penult r correspondin r	re premium ba oth A- and H- oxies of differt ibles in all mo inate past six ig heterosked	sed on proxy of share markets; t snces of opinion dels are market -month return (asticity-consiste	differences of the dependent s <i>SIGMA</i> and c aptitalization <i>RET712</i>); the int <i>t</i> -statistics
	Re A-sh diffe and									Ch
Ň	Tal gression re are price pr based on p rences of o d control va									Dual l inese sl
www.	ble IV. sults of remium roxy of pinions ariables								75	isted nares

es

ons les

return momentums (*RETs*) do influence the magnitude of A-share price premiums. Specifically, the A-share market capitalization, the H-share book-to-market ratio and the past A-share returns are positively related to the A-share premiums; while the H-share market capitalization, the A-share book-to-market ratio and the past H-share returns are negatively related to the A-share premiums. Compared with our main variables of interest *SIGMA* and *TURN* in Models 1 and 2, the control variables also have substantial explanatory power to the A-share premiums with the adjusted R^2 ranges from 10.24 percent (*RET1*) to 45.47 percent (*MV*).

In the regression Models 8 and 9, we use all differences of opinions proxy variables and control variables to test whether the effects on A-share premiums generated by differences of opinions proxy variables are distinct from that generated by control variables. The results shown in the last two columns of Table IV indicate that the proxies of differences of opinions *SIGMA* and *TURN* have the incremental explanatory power on the A-share premiums. The coefficients of *SIGMA* and *TURN* are all significantly positive in two models. For example, the coefficient of *SIGMA* ^A is 5.29 with *t*-statistic 6.56 in Model 8. The result implies that 1 percent increase in A-share idiosyncratic return volatility will lead to a 5.29 percent increase in A-share price premium, *ceteris paribus*. The implications of other coefficients of differences of opinions are similar.

In addition, the estimated coefficients of all control variables in Models 8 and 9 have the predicted sign and are significant at 5 percent levels or above except for the H-share BM ratio and past penultimate six-month return (*RET712*). The regression results indicate that both proxies of differences of opinions and prevailing pricing factors have explanatory power on A-share price premiums. The weak significance of coefficients of *RET712*'s implies that the investment horizon of most investors is no longer than six months in both A- and H-share markets. It seems that investors in both markets like to pursue stocks with recent high abnormal returns regardless of their fundamental values. Such investment behavior is relatively easy to cause speculative bubble in the A-share market as indicated by Mei *et al.* (2003). Besides, the speculative investors are more likely to be affected by the differences of opinions than the value investors. The regression results in Table IV support our *H1* and *H2* that high differences of opinions lead to an upward bias in A-share prices and cause large price dispersions in the segmented A- and H-share markets with stringent short-sale constraint.

4.4 Robustness regression analysis with alternative proxies of differences of opinions

In this section we use robustness regression test to study whether the relationships between A-share price premiums and proxies for differences of opinions derived from the main regression model (equation (8)) in Table IV remain valid when we use alternative proxy of differences of opinions and control for other factors that may have explanatory power over A-share premiums.

In order to rule out the possibility that our results on A-share price premiums are mainly driven by *SIGMA* and *TURN* in the main regression model of equation (8), we use one alternative proxy *RETVOL* for the differences of opinions. The regression result using *RETVOL* for the differences of opinions is shown in the first model (column one) of Table V. The result shows that the coefficients of both A- and H-share *RETVOLs* are positive and significant at 5 percent level or above. The result implies that 1 percent increase in A-share (H-share) monthly excess return volatility will lead to a 3.74 percent (1.50 percent) increase in A-share price premium. The result on *RETVOL* provides further



CFRI

3.1

	1 (RETVOL)	2 (SIGMA)	3(SIGMA)	4(TURN)	5 (<i>TURN</i>)	Dual listed Chinese shares
DIFF_a	3.74 ^{***}	5.32 ^{***} (6.37)	1.28 ^{***} (2.96)	0.17^{***}	0.03*	Chinese shares
DIFF_h	1.50**	0.67	0.55	0.33***	0.13***	
MV_a	(2.55) 12.37***	(0.89) 12.07***	(1.15) 22.33 ^{***}	(8.44) 11.41***	(3.34) 21.53 ^{***}	77
MV_h	(10.78) - 17.51 ***	(10.69) -17.20^{***}	(5.63) - 25.29 ***	(9.39) - 17.52***	(5.41) - 24.21 ***	
BM_a	(-11.92) -10.81^{***}	$(-12.41) - 7.23^*$	(-11.36) - 3.33	(-11.73) -13.81***	(-11.21) - 3.25	
BM_h	(-3.01) -0.01	(-1.93) 0.75	(-1.45) 4.86^{***}	(-3.49) 3.40 (1.20)	(-1.49) 4.28^{***}	
RET1_a	(-0.00) 0.19^{***}	(0.30) 0.18 ^{**}	(2.71) 0.07***	(1.29) 0.15*	(2.65) 0.07***	
RET1_h	(3.19) -0.19^{***}	(2.45) -0.17^{***}	(5.11) -0.15^{***}	(1.88) -0.14^{**}	(4.70) -0.17^{***}	
RET26_a	(-3.48) 0.12^{***}	(-2.99) 0.10**	(-7.70) 0.05^{***}	(-2.15) 0.12^{***}	(-8.00) 0.06***	
RET26_h	(3.64) - 0.06 **	(2.59) - 0.06*	(3.51) - 0.07***	(3.15) -0.13^{***}	(3.42) - 0.08 ***	
RET712_a	(-2.32) 0.06^{**}	(-1.74) 0.03	(-4.25) 0.04^{***}	(-4.25) 0.31	(-5.43) 0.04^{***}	
RET712_h	(2.00) - 0.03	(0.85) - 0.01	(3.07) -0.03^{**}	(0.82) - 0.03	(3.45) -0.03^{***}	
Beta_a	(-1.19)	(-0.16) 2.50 (1.54)	(-2.11) -1.71 (-0.48)	(-0.92) 0.65 (0.36)	(-3.17) -1.07 (-0.30)	
Beta_h		1.64	(-5.92^{***})	(0.30) -1.20 (-0.87)	(-5.72^{***})	
MKTRET_a		(1.22)	0.13***	(-0.87)	0.13***	
MKTRET_h			(5.01) -0.27^{***}		(5.07) -0.32^{***}	
EXP			(-9.85) 1.52*** (8.42)		(- 10.58) 1.34*** (7.25)	
Firm effect	No	No	(8.43) Yes	No	(7.25) Yes	
Adjusted R^2 (%)	56.76	56.77	res 53.73	1NO 60.88	res 46.70	

Notes: Significant at: *10, **5 and ***1 percent levels; this table reports the robustness regression results of A-share price premium with one alternative proxy of differences of opinions, *RETVOL* and also controlling for the risk factor, market condition and expected RMB exchange rate change; *RETVOL* is the monthly excess return volatility; the dependent variable is the price premium of A-shares, *PERM*; the alternative proxy *RETVOL* and two main proxies *SIGMA* and *TURN* in the robustness regression models are specified by *DIFF* (*DIFF_a* and *DIFF_h*) in A- and H-share markets, respectively; the table shows the regression estimate of weighted Fama and Macbeth coefficients and their corresponding heteroskedasticity-consistent *t*-statistics (in parenthesis), where the weights correspond to the number of observations in each month without control of firm effect and time effect (Models 1, 2 and 4); the table also shows the regression estimate of panel regression with control of both firm effect and time effect (Models 3 and 5)

لاستشارات

Table V.

Robustness regression results of A-share price premium with alternative proxy of differences of opinions support to our first two hypotheses that the absolute magnitude of differences of opinions in both A- and H-share markets generates positive impacts to the A-share price premiums. Of particular importance, we test whether the price premiums of A-shares are caused by the contemporaneous risk factors embedded in the pricing of the dual-listed stocks. We estimate *beta* from equation (4) where *beta* = $\beta_1 + \beta_2$. The predicted sign of the beta coefficient is unknown, depending on the investors' attitude toward risk. Specifically, if investors of both A- and H-share markets are risk lover, we expect that the high A-share beta or the low H-share beta would lead to the large price premiums of A-shares. Thus, the coefficient of A-share beta should be positive and that of H-share beta should be negative. On the other hand, if investors in both markets are risk averse, then the coefficient of A-share beta is expected to be negative and that of H-share beta is expected to be positive.

In Models 2 and 4 of Table V, we add betas into the main regression model of equation (8) where the proxies of differences of opinions are still *SIGMA* and *TURN*. The results indicate that the risk factors have nearly no impacts on our conclusions regarding the effects of differences of opinions on the A-share premiums. The magnitude and significance of the coefficients for the differences of opinions proxies are quite similar to those estimated in the main regression models in Table IV. For example, the coefficients of A- and H-share *SIGMA* in the regressions with risk factors are 5.32 and 0.67, respectively, very close to that of 5.29 and 1.10 in the main regression of Model 8 in Table IV. In particular, three of the four coefficients of the differences of opinions are significant at 5 percent level or above in the regressions with risk factors. However, all four beta coefficients are insignificant. The results indicate that the effects of risk factors on the A-share price premiums are at least partially subsumed to that of the differences of opinions or other prevailing pricing factors.

In Models 3 and 5 of Table V, we further add the market return and exchange rate change as the control variables to study the effect of *SIGMA* and *TURN* on the price premiums of A-shares[3]. In these two models, the coefficients of A-share market return (*MKTRET*^A) and RMB expected exchange rate change (*EXP*) are positive and significant at 1 percent level while the coefficient of H-share market return (*MKTRET*^H) is negative and also significant at 1 percent level. The signs of these coefficients are consistent with the existing literature. The results indicate that when the investment sentiment and market return of A-shares are high and investors expect RMB tend to be appreciated, the A-share premiums will increase significantly. However, when the investment sentiment and market return of H-share are high, the A-share premiums will be expected to decline. Moreover, even after we control for risk factor, market condition and the expected change of RMB exchange rate, we still find that the differences of opinions *SIGMA* and *TURN* have incremental explanatory power for the A-share premiums. Specifically, three of the four coefficients of *SIGMA* and *TURN* are positive and significant at 10 percent level or above in Models 3 and 5.

4.5 Event study of A-share price premiums with relaxation of short-sale constraint

On March 20, 2010, CSRC introduced the short-sale mechanism into the A-share market. In this section, we study the impacts of newly launched short-sale policy on the A-H share price dispersions. Specifically, we test our *H3* whether the new short-sale policy in A-share market will facilitate the price discovery of A-shares and cause the price dispersions of shorted A-shares to be narrowed more than those of non-shorted ones.



CFRI

3.1

4.5.1 Portfolio analysis on joint effects. For the event study, our test period is from March 2009 to March 2011. To explore the joint effects of new short-sale policy and differences of opinions on A-share price premiums, we first categorize all sample A-shares into two portfolios as "shorted" and "non-shorted" portfolios according to the new short-sale policy. We then divide the sample into two sub-periods for each A-share portfolio. The "pre-event" sub-period is from March 2009 to March 2010 while the "post-event" sub-period is from April 2010 to March 2011. Finally, we further categorize the shorted and non-shorted A-share portfolios into two sub-portfolios, respectively, based on the high or low magnitude of SIGMA, TURN, r SIGMA and r TURN. For A-share portfolios constructed above, we first calculate the mean price premium of shorted and non-shorted A-share portfolios in both pre- and post-event sub-periods, respectively, based on the high or low level of SIGMA, TURN, r SIGMA and r_TURN . Then, the A-share price premium change between pre- and post-event sub-period is calculated for a given level of SIGMA, TURN, r_SIGMA and r_TURN. Our ultimate goal of the portfolio analysis of event study is to test whether the price premium change of shorted A-share portfolio is larger than that of non-shorted ones for each level of SIGMA, TURN, r_SIGMA and r_TURN. The differences of A-share price premium change between pre- and post-event sub-period of shorted and non-shorted A-share portfolios are shown in the last column of Table VI named "difference in difference".

Panel A of Table VI reports the mean price premium of A-share portfolios based on the absolute level of *SIGMA*. The results shown in the last column support our *H3* that the new policy causes the price premium of shorted A-share portfolio to decline more than that of non-shorted portfolio. For example, when A-share *SIGMA* is low, the price premium of shorted A-share portfolio drops 10.58 percent more than that of non-shorted portfolio. The more significant result to support our *H3* is that the mean price premium of shorted A-share portfolio in the post-event period is only -0.54 percent and insignificant. The price premium is gone! The results based on H-share *SIGMA* are also similar. When *SIGMA* of H-shares is low, the price premium of shorted A-share portfolio is also almost disappeared in the post-event period. The premium change comparison of the last column shows that the price premium of shorted A-share portfolio drops 11 percent more than that of non-shorted portfolio on average.

Panel B of Table VI reports the mean price premium of A-share portfolios based on the absolute level of *TURN*. Consistent with the results shown in Panel A, the price premium of shorted A-share portfolio is eliminated in the post-event period when *TURN*s of either A- or H-share portfolios are low. As can be seen from the last column for premium change comparisons, the price premium of shorted A-share portfolio drops more than that of non-shorted portfolio in three out of four cases. The price premium of shorted A-share portfolio drops 5.95 percent (with *t*-statistic of 2.60) and 18.06 percent (with *t*-statistic of 10.59), respectively, for the case of high and low H-share turnover rates and drops 12.72 percent (with *t*-statistic of 5.52) more for the case of low A-share turnover rate. The only exception is for the case of high A-share turnover rate in which the difference of price premium change is only insignificantly -0.81 percent. Overall, the mean price premium of A-share portfolios based on *TURN* also supports our *H3*. Figure 3 shows that the A-share price premium of shorted A-share portfolios has dropped much sharper than that of non-shorted ones in the same period after the relaxation of short-sale constraint in the A-share market.



CI	F)	R	
3.	1		

80

Table VI. Event study of A-share price premium with the relaxation of short-sale constraint

المتسارات

		Shor Pre-event	ted A-share portfolio Post-event	Post-pre	Non-s Pre-event	horted A-share portfc Post-event	dio Post-pre	Difference in difference
Panel A: mean SIGMA_a	: price premium Low	of A-share portfolio 0.2043 *** (9.37) 0.237)	(-0.0054 (-0.23)	-0.2096 *** (-8.17)	0.4416*** (22.65) 0.2200.***	0.3377 * * * (17.08)	-0.1038^{***} (-4.32)	0.1058 * * * (3.40)
SIGMA_h	High Low	$\begin{array}{c} 0.4236\\ (13.99)\\ 0.2221 ^{***}\\ (12.38) \end{array}$	0.2335 (12.60) 0.0232 (1.03) ****	$egin{array}{c} - 0.1701 \ (-5.01) \ - 0.2069 & ** & * \ - 12.81) \ - 12.81) \ & ** & * & * \end{array}$	$\begin{array}{c} 0.5772\\ (39.33)\\ 0.4905 ^{***}\\ (20.06) & & & & & & & & & & & & & & & & & & &$	$\begin{array}{c} 0.4583\\ (33.56)\\ 0.3979 ^{***}\\ (24.33)\\ 24.33 \end{array}$	-0.1189 (-7.69) -0.0969 *** (-3.56)	$\begin{array}{c} 0.0513\\ (1.84)\\ 0.1100 & * & * \\ (3.62) & * & * \end{array}$
Danal R. maan	High Arrico Arcomium	$\begin{array}{c} 0.4405\\ (15.87)\\ of A \ e \ barrow \ have \ for \ follow \end{array}$	0.2205	-0.2200 (-5.82)	(45.09)	0.4222 (33.23)	-0.1170 (-15.30)	0.1030 (2.45)
TURN_a TURN_a	price premuum Low High	0) A-Share poryoud 0.2757 *** (14.43) 0.2970 ***	$\begin{array}{c} 0.0439 \\ (1.68) \\ 0.1796 * * * \end{array} (-$	-0.2318^{***} -14.21) -0.1174^{***}	$\begin{array}{c} 0.4949^{***} \\ (29.03) \\ 0.5496^{***} \end{array}$	0.3903 * * * (37.04) 0.4240 * * *	$\begin{array}{c} -0.1046^{***} \\ (-6.84) \\ -0.1256^{***} \end{array}$	0.1272^{***} (5.52) -0.0081
TURN_h	Low High	(12.01) 0.1962^{***} (10.80) 0.4267^{***}	$\begin{array}{c} (8.82) \\ -0.0344 \\ (-1.63) \\ 0.2123 \\ * * * \end{array} (-$	(-3.13) -0.2305 *** -15.98) -0.2143 ***	(34.33) 0.4632 *** (19.55) 0.5609 ***	(31.74) 0.4132 *** (30.50) 0.4061 *** (21.22)	(-12.21) -0.0499 ** (-2.22) -0.1548 ***	(-0.21) 0.1806 *** (10.59) 0.0595 **
Panel C: mean r_SIGMA	<i>price premium</i> Low Hiah	$of A-share portfolio 0.2772^{***}_{0.260}$ (14.36) 0.2939^{***}_{0.2939}	(1.0.01) $(0.1002^{***}$ (3.86) $(0.084d^{***}$	(-0.00) $(-0.1770^{***}$ (-5.15) $(-0.2094^{***}$	(120.96) (29.96) (29.1 ***	(29.78) (29.78) (239.4 * * *)	-0.1022 *** (-6.29) (-6.29) (-0.1207 ***)	$\begin{array}{c} (0.0748 * * \\ (2.04) \\ (0.0888 * * * * \end{array} \end{array}$
$r_{-}TURN$	Low High	(10.25) 0.3350 *** (14.18) 0.2560 *** (11.04)	(3.65) 0.0992 *** (5.15) 0.0854 *** (4.38)	-1054) -0.2358*** -11.78) -0.1706*** (-6.24)	(27.00) 0.5449 *** (33.99) 0.4876 *** (27.51)	(29.29) 0.3656 *** (21.84) 0.4418 *** (41.02)	(-6.93) -0.1793 *** (-10.40) -0.0458 ** (-2.68)	(4.15) 0.0565 ** (2.38) 0.1248 ***
Notes: Signifi and after the r "shorted" and ' March 2009 to	cant at: *10, ** elaxation of str 'non-shorted" po March 2010 whi	⁵ ⁵ and ^{***} ¹ percent ict short-sale constra ortfolios according to ile the "post-event" st	levels; this table reports init; the test period of ev the new short-sale polic ub-period is from April 2	the event study resu the study is from M y; we then divide the 2010 to March 2011; f	llts of price premium larch 2009 to March sample into two sub inally, we further ca	1 changes between shu 2011; we first catego -periods for each A-sh tegorize the shorted a	orted and non-shorted wrize all sample A-shar tare portfolio; the "pre- nd non-shorted A-shar	A-share portfolios before tes into two portfolios as event" sub-period is from e portfolios into two sub-

portfolios, respectively, based on the low or high magnitude of SIGMA, TURN, r_{-SIGMA} and $r_{-}TURN$; we calculate the mean price premium of shorted and non-shorted A-share portfolios in both pre- and post-event sub-periods based on SIGMA, TURN, $r_{-}SIGMA$ and $r_{-}TURN$ panels A-C report the mean price premium of A-share portfolio based on SIGMA, TURN and $r_{-}SIGMA$ and $r_{-}TURN$ of A- and H-shares, respectively, and its corresponding heteroskedasticiy-consistent *t*-statistics (in the parentheses); the difference of the premium change of the A-share portfolio between pre- and post-event sub-period is evaluated with a two-sided *t*-test and presented in the column as "difference" in difference.



Panel C of Table VI reports the mean price premium of A-share portfolios based on the relative magnitude of differences of opinions r_SIGMA and r_TURN . The results shown in the last column support our H3 that the new policy caused the price premium of shorted A-share portfolios to decline more than that of non-shorted ones. For example, when r_SIGMA is low, the A-share price premium of shorted A-share portfolio. And when r_SIGMA is high, the difference of A-share price premium between shorted and non-shorted A-share portfolios becomes 8.88 percent (with *t*-statistic of 4.15). The similar results are obtained for the study of r_TURN . The difference of mean price premium between shorted and non-shorted and non-shorted A-share portfolios increases from 5.65 percent (with *t*-statistic of 2.38) when r_TURN is low to 12.48 percent (with *t*-statistic of 3.98) when r_TURN is high.

4.5.2 Regression analysis on joint effects. To further investigate the movement of A-share price premium after March 2010, we use the regression analysis to study the joint effects of new short-sale policy and differences of opinions on the A-share price premium. Since the length of test period is only 13 months from March 2010 to March 2011, we only include the market capitalization *MV* and book-to-market ratio *BM* in A- and H-share markets as the control variables along with the shorted dummy, interaction terms and proxies of differences of opinions in the joint effects regression model:

$$PREM_{i,t} = \beta_0 + \beta_1 DIFF_{i,t}^A + \beta_2 DIFF_{i,t}^H + \beta_3 SHORT_i + \beta_4 SHORT_i * DIFF_{i,t}^A$$

$$+ \beta_5 SHORT_i * DIFF_{i,t}^H + \beta_6 MV_{i,t}^A + \beta_7 MV_{i,t}^H + \beta_8 BM_{i,t}^A + \beta_9 BM_{i,t}^H + \mu_{i,t}$$
(9)

where *PREM* is the mean A-share price premium, *DIFF*^A and *DIFF*^H are the absolute level of proxies of differences of opinions (*SIGMA*, *TURN* and *RETVOL*) in A- and H-share markets, respectively, *SHORT* is a dummy variable which equals one if A-share can be shorted according to the new short-sale policy and zero otherwise, *SHORT***DIFF*^A and *SHORT***DIFF*^H are the interaction terms of short-sale constraint and differences of opinions in A- and H-share markets, respectively, MV^A and MV^H are the market capitalization and BM^A and BM^H are the book-to-market ratio in A- and H-share markets, respectively. Table VII reports the regression results of joint effects of new short-sale policy and differences of opinions on A-share price premium.



82

			3			6
	1 (SIGMA)	2 (<i>TURN</i>)	(RETVOL)	4 (SIGMA)	5 (<i>TURN</i>)	(RETVOL)
DIFF_a	7.24 ***	0.06	4.15***			
DIFF_h	(5.60) - 1.76	(1.40) -0.02	(3.98) 0.41			
r_DIFF	(-1.09)	(-0.11)	(0.27)	7.66 ^{***} (3.31)	-3.05^{***}	5.35 [*] (1.68)
SHORT	-22.35***	-24.98***	- 32.60 ***	-4.94***	-2.54***	-4.71***
SHORT*DIFF_a	(-5.32) 10.98 ^{**}	(-7.01) 0.24^{***}	(-6.20) 12.84 ^{***}	(-4.72)	(-3.77)	(-4.31)
SHORT*DIFF_h	(2.37) 1.69	(3.35) 1.66***	(3.92) -0.12			
SHORT*r_DIFF	(0.55)	(0.15)	(-0.04)	5.99	2.40 ***	18.29***
MV_a	18.51***	16.43***	18.35 ***	(1.55) 17.46***	(5.15) 13.67***	(2.64) 18.32***
MV_h	(13.24) -24.49***	(19.00) -23.23^{***}	(12.42) -24.26***	(18.78) -25.99^{***}	(15.79) -24.06***	(19.27) - 26.25 ^{***}
BM_a	(-19.90) 7.94***	(-48.85) 7.87***	(-19.24) 8.18***	(-35.03) 7.12^{***}	(-38.02) 6.13^{***}	(-33.99) 7.79***
BM_h	(8.95) 1.28	(12.14) 1.44	(8.83) 2.31	(11.47) - 2.94 ^{**}	(14.13) -5.02***	(11.16) - 2.97***
Adjusted R^2 (%)	(0.85) 57.97	(0.66) 56.08	(1.25) 59.09	(-2.02) 56.76	(-4.40) 53.83	(-2.19) 56.33

Notes: Significant at: *10, **5 and ***1 percent levels; this table provides the regression results of A-share price premium based on the joint effects of short-sale constraints and differences of opinions after the relaxation of short-sale constraint in the A-share market; the sample period is from March 2010 to March 2011; the dependent variable in all models is the mean price premium of A-shares, *PREM*; the absolute level of proxy of differences of opinions in the first three models is specified by *DIFF* in both A- and H-share markets (*DIFF_a* and *DIFF_h*), respectively; the relative level of proxy of differences of opinions in Models 4-6 is specified by *r_DIFF* in A- and H-share markets, respectively; *SHORT* is the shorted dummy variable; *SHORT*DIFF_a* and *SHORT*DIFF_h* are the interaction terms of short-sale constraint and absolute level of differences of opinions in A- and H-share markets, respectively; *SHORT*r_DIFF* is the interaction term of short-sale constraint and absolute level of differences of opinions in A- and H-share markets, respectively; *SHORT*r_DIFF_h* are the interaction term of short-sale constraint and relative level of differences of opinions in A-H share markets; the control variables in all models are the market capitalization $MV(MV_a \text{ and } MV_h)$ and book-to-market ratio $BM(BM_a \text{ and } BM_h)$, respectively, in both A- and H-share markets; the table shows the regression estimate of weighted Fama and Macbeth coefficients and their corresponding heteroskedasticity-consistent *t*-statistics (in parenthesis), where the weights correspond to the number of observations in each month

Regression results of A-share price premium based on the joint effects of short-sale constraints and differences of opinions

Table VII.

In the first three models of Table VII, we test the joint effects of absolute level of differences of opinions and short-sale constraints on the A-share premium. While in Models 4-6 of Table VII, the joint effects of relative level of differences of opinions and short-sale constraints on the A-share premium are examined[4]. In the regression results of Table VII, the coefficients of shorted dummy variables are our main interest. As shown in the first three columns of Table VII, when the absolute level of differences of opinions is used in the test, the coefficients of shorted dummy in all three models are significantly negative. They are -22.35(with *t*-statistic of -5.32), -24.98 (with *t*-statistic of -7.01) and -32.60 (with *t*-statistic of -6.20), respectively, when *SIGMA*,



TURN and *RETVOL* are used as proxies of difference of opinions. The results indicate that the relaxation of short-sale constraint in the new policy has greatly narrowed the price premium of shorted A-share portfolio, which further supports our *H3*.

The results with respect to the main effects of differences of opinions are mixed. The coefficients of proxy of differences of opinions in both A- and H-share markets are not as remarkable as they are in the main and robustness regression tests (Tables IV and V). Only the coefficients of *SIGMA*^A and *RETVOL*^A have significant positive values of 7.24 (with *t*-statistic of 5.60) and 4.15 (with *t*-statistic of 3.98), respectively. Other coefficients of *SIGMA*^H and *RETVOL*^H in the H-share market and *TURN*^A and *TURN*^H in both A- and H-share markets are statistically insignificant. The interaction effects between short-sale constraints and differences of opinions are positive and significant at 5 percent level reflected in four of the six coefficients. The exceptions are the coefficients of *SIGMA*^H and *RETVOL*^H which is 1.69 (with *t*-statistic of 0.55) and -0.12 (with *t*-statistic of -0.04), respectively. The interaction results indicate that with the control of short-sale constraint, the higher the idiosyncratic return volatility or excess return volatility in A-share market and the monthly turnover rate in both A- and H-share markets, the larger will be the A-share price premium.

We obtain qualitatively similar results in Models 4-6 (last three columns) of Table VII when the joint effects of relative level of differences of opinions and short-sale constraints are examined. As can be seen, the coefficients of shorted dummy in Models 4-6 are all significantly negative but with smaller magnitude compared with that in the first three columns. They are -4.94 (with *t*-statistic of -4.72), -2.54 (with *t*-statistic of -3.77) and -4.71 (with *t*-statistic of -4.31), respectively, when *SIGMA*, *TURN* and *RETVOL* are used as proxy of difference of opinions. Compared with the first three models, the coefficients of relative level of differences of opinions are all significant, particular for *r_SIGMA* which is 7.66 (with *t*-statistic of 3.31) and r_TURN which is -3.05 (with *t*-statistic of -6.54), respectively. The interaction effects between short-sale constraints and relative level of differences of opinions are positive and significant at 1 percent level reflected in two of the three coefficients.

The main effects of differences of opinions in the regression tests are consistent with the findings of Boehme et al. (2006) that the stocks with significant overvaluations are subjected to both differences of opinions and short-sale constraints. Stocks are not systematically overvalued when either one condition is not met. In addition, the first two models of Table VII show that the coefficients of market capitalization MV are significantly positive at 18.51 (with t-statistic of 13.24) and 16.43 (with t-statistic of 19.00) in the A-share market and significantly negative at 24.49 (with t-statistic of 19.90) and 23.23 (with t-statistic of 48.85) in the H-share market, when SIGMA and TURN are used as the differences of opinions, respectively. The coefficients of book-to-market ratio BM are significantly positive at 7.94 (with t-statistic of 8.95) and 7.87 (with t-statistic of 12.14) in the A-share market and statistically insignificant in the H-share market, when SIGMA and TURN are used as the differences of opinions, respectively. The regression results of Table VII indicate that market capitalization MV and book-to-market ratio BM of A-shares generate significant positive effect to the A-share price premium. However, the market capitalization MV of H-shares generates significant negative effect to the A-share price premium. The results of Table VII are consistent with the patterns we documented in the early analysis.



CFRI 3,1

84

5. Summary and conclusions

In this study, we provides evidence that dual-listed Chinese A-shares with high level of differences of opinions and short-sale constraints tend to be overvalued and the price premium between dual-listed A-H shares tends to be significant. Our analysis mainly follows the Miller's (1977) model, which indicates that the relaxation of stringent short-sale constraint could reduce the upward bias in stock price.

Following the literature, we use idiosyncratic return volatility and monthly turnover rate as two main proxies of differences of opinions. The relaxation of stringent short-sale constraint in A-share market in March 2010 provides us an opportunity to analyze the joint effects of short-sale constraints and differences of opinions on the A-share price premium. Our results indicate that with strict short-sale constraints, the high level of differences of opinions in both A- and H-share markets will lead to the high price premium of A-shares. The results prevail even if we control for the pricing factors such as the market capitalization, book-to-market ratio, past return momentums and other potential pricing factors. The effects of differences of opinions on dual-listed A-shares are quite distinctive. It seems a bit counter-intuitive that the level of H-share differences of opinions is positively related to the A-share price premium. This is due to the different institutional setting between mainland China and Hong Kong stock markets. As Hong Kong stock market is very open and internationalized, both optimistic and pessimistic opinions are reflected in H-share prices. Thus, the differences of opinions in H-share market may not cause the H-share prices to deviate from their fundamental values. However, the A-share market tends to be more domestic oriented and overvalued. As a result, the high level of differences of opinions in H-share market could also lead to the large price premium of A-shares.

The event study shows that after the introduction of short-sale mechanism in A-share market, the price premium of shorted A-share portfolio drops more significantly compared with that of non-shorted ones. The significant negative coefficients of shorted dummy variables in the regression analysis explain why the price premium of shorted A-share portfolio is narrowed with the relaxation of short-sale constraint in the A-share market. The positive regression coefficients of the interaction terms support Miller's hypothesis that with the control of short-sale constraints, the high level of differences of opinions could lead to the high degree of overvaluation of A-share prices. The regression results also indicate that market capitalization and book-to-market ratio of A-shares generate significant positive effect to the A-share price premium that are consistent with the literature. Our study provides further evidence that the introduction of short-sale mechanism in A-share market could partially eliminate the mispricing of dual-listed A-shares and improve the price efficiency of A-share market.

Notes

- 1. The descriptive information of firms in our study with company name and sector, the stock code and IPO date in both A- and H-share markets is presented in Appendix 1.
- 2. Appendix 2 provides the list of firms with shorted A-shares after the introduction of short-sale policy in A-share market. As they are all large capitalization stocks, the corresponding H-shares are likely to be not subjected to the short-sale constraints in the sample period.



- 3. We use panel regression method adopted by Arquette *et al.* (2008) in these two models. In fact, the results of Models 2 and 4 in Table V are not the same for the Fama-MacBeth regression and panel regression with beta coefficients insignificant in Fama-MacBeth regression and significant in panel regression.
- 4. The following regression model is used to obtain the results of Models 4-6 in Table VII:

$$PREM_{i,t} = \beta_0 + \beta_1 r_DIFF_{i,t} + \beta_2 SHORT_i + \beta_3 SHORT_i * r_DIFF_{i,t} + \beta_4 MV_{i,t}^A$$

$$\vdash \beta_5 M V_{i,t}^H + \beta_6 B M_{i,t}^A + \beta_7 B M_{i,t}^H + \mu_{i,t}$$

where r_DIFF is the relative level of proxy of differences of opinions (r_SIGMA , r_TURN and r_RETVOL) in A- and H-share markets, respectively, and other variables are the same as that in equation (9).

References

- Arquette, G.C., Brown, W.O. and Burdekin, R.C.K. (2008), "US ADR and Hong Kong H-share discounts of Shanghai-listed firms", *Journal of Banking & Finance*, Vol. 32 No. 9, pp. 1916-27.
- Bailey, W. and Jagtiani, J.A. (1994), "Foreign ownership restrictions and stock prices in the Thai capital market", *Journal of Financial Economics*, Vol. 36 No. 1, pp. 57-87.
- Berkman, H., Dimitrov, V., Jain, P.C., Koch, P.D. and Tice, S. (2009), "Sell on the news: differences of opinion, short-sale constraints and returns around earnings announcements", *Journal of Financial Economics*, Vol. 92 No. 3, pp. 376-99.
- Boehme, R.D., Danielsen, B.R. and Sorescu, S.M. (2006), "Short-sale constraints, differences of opinion and overvaluation", *Journal of Financial and Qualitative Analysis*, Vol. 41 No. 2, pp. 455-87.
- Burdekin, R.C.K. and Redfern, L. (2009), "Sentiment effects on Chinese share prices and saving deposits: the post-2003 experience", *China Economic Review*, Vol. 20 No. 2, pp. 246-61.
- Cai, C.X., McGuinness, P.B. and Zhang, Q. (2011), "The pricing dynamics of cross-listed securities: the case of Chinese A- and H-shares", *Journal of Banking & Finance*, Vol. 35 No. 8, pp. 2123-36.
- Callen, J.L., Lai, K. and Wei, S.X. (2009), "Understanding the variation of foreign share price discounts: a study of dual-listed Chinese firms", working paper, Hong Kong Polytechnic University, Hong Kong.
- Chan, K., Kot, H.W. and Yang, Z.S. (2010), "Effects of short-sale constraints on stock prices and trading activity: evidence from Hong Kong and Mainland China", working paper, Hong Kong University of Science and Technology, Hong Kong.
- Chan, K., Menkveld, A.J. and Yang, Z.S. (2008), "Information asymmetry and asset prices: evidence from the China foreign share discount", *Journal of Finance*, Vol. 63 No. 1, pp. 159-96.
- Chen, J., Hong, H. and Stein, J.C. (2002), "Breadth of ownership and stock returns", Journal of Financial Economics, Vol. 66 Nos 2/3, pp. 171-205.
- Chong, T.T.L. and Su, Q. (2006), "On the comovement of A and H shares", *Chinese Economy*, Vol. 39 No. 5, pp. 68-86.
- Diether, K.B., Malloy, C.J. and Scherbina, A. (2002), "Difference of opinion and the cross section of stock returns", *Journal of Finance*, Vol. 57 No. 5, pp. 2113-41.
- Fama, E.F. and French, K.R. (1992), "The cross-section of expected stock returns", Journal of Finance, Vol. 47 No. 2, pp. 427-65.



Dual listed Chinese shares

(10)

85

CFRI 3 1	Fama, E.F. and MacBeth, J. (1973), "Risk, return and equilibrium: empirical tests", Journal of Political Economy, Vol. 81 No. 3, pp. 607-36.
0,1	Johnson, T.C. (2004), "Forecast dispersion and cross section of expected returns", <i>Journal of Finance</i> , Vol. 59 No. 5, pp. 1957-78.
	Karolyi, G.A. and Li, L.F. (2002), "A resolution of the Chinese discount puzzle", working paper, Ohio State University, Athens.
86	Li, Y.M., Yan, D.Y. and Greco, J. (2006), "Market segmentation and price differentials between A shares and H shares in the Chinese stock markets", <i>Journal of Multinational Financial</i> <i>Management</i> , Vol. 16 No. 3, pp. 232-48.
	Mei, J.P., Scheinkman, J.A. and Xiong, W. (2003), "Speculative trading and stock prices: an analysis of Chinese A-B share premia", working paper, New York University, New York, NY.
	Miller, E. (1977), "Risk, uncertainty and divergence of opinion", <i>Journal of Finance</i> , Vol. 32 No. 4, pp. 1151-68.
	Nagel, S. (2005), "Short sales, institutional investors and the cross-section of stock returns", Journal of Financial Economics, Vol. 78 No. 2, pp. 277-309.
	Sun, Q. and Tong, W.H.S. (2000), "The effect of market segmentation on stock prices: the China syndrome", <i>Journal of Banking & Finance</i> , Vol. 24 No. 12, pp. 1875-902.
	Tang, V.W. (2011), "Isolating the effect of disclosure on information risk", <i>Journal of Accounting and Economics</i> , Vol. 52 No. 1, pp. 81-99.
	Tong, W.H.S. and Yu, W.W. (2012), "A corporate governance explanation of the A-B share discount in China", <i>Journal of International Money and Finance</i> , Vol. 31 No. 2, pp. 125-47.
	Wang, S.S.Y. and Jiang, L. (2004), "Location of trade, ownership restrictions and market illiquidity: examining Chinese A- and H-shares", <i>Journal of Banking & Finance</i> , Vol. 28 No. 6, pp. 1273-97.
	White, H. (1980), "A heteroskedasticity-consistent covariance matrix estimation and a direct test for heteroskedasticity", <i>Econometrica</i> , Vol. 48 No. 4, pp. 817-38.
	About the authors Dr Zhenmin Fang is currently an Associate Professor of Finance in the Department of Economics and Finance, City University of Hong Kong. He received the Master of Applied Mathematics from Huazhong University of Science and Technology, China in 1983 and PhD in Business (Finance) from Baruch College, City University of New York in 1990. His main research interest is in the area of portfolio choices and investment, futures and derivative securities and China stock markets. Zhenmin Fang is the corresponding author and can be contacted at: efjimmy@cityu.edu.hk Mr Xin Jiang is currently a Doctoral Student in the Department of Economics and Finance
	City University of Hong Kong He restricted the Bachelor of Economics from Naplesi University in

City University of Hong Kong. He received the Bachelor of Economics from Nankai University in 2010. His main research interest is in the area of empirical corporate finance and market microstructure.

To purchase reprints of this article please e-mail: **reprints@emeraldinsight.com** Or visit our web site for further details: **www.emeraldinsight.com/reprints**



Appendix 1	Dual listed Chinese shares
Sector Endotrials Public utility Finance Public utility Industrials Public utility Finance Public utility Public utility Public utility Public utility Industrials	Public utility Public utility Public utility Conglomerates (continued)
H-share IPO date January 21, 1998 November 13, 1996 November 13, 1996 November 11, 1994 June 30, 1999 June 30, 1997 June 12, 1997 June 12, 1997 June 12, 1997 June 12, 1997 June 12, 1997 June 27, 1993 March 12, 1993 March 29, 1994 July 26, 1993 May 17, 1994 June 6, 1993 May 17, 1994 June 6, 1993	October 17, 1997 June 15, 2005 October 7, 1997 March 13, 2008 July 16, 2010
A-share IPO date December 6, 2001 January 7, 2003 December 19, 2000 May 23, 2005 February 3, 2005 April 25, 2003 April 9, 2002 July 1, 1998 February 6, 2001 January 11, 2002 January 11, 2002 January 11, 2002 January 11, 2002 January 11, 2002 January 6, 1994 May 6, 1994 May 6, 1994 May 6, 1994 May 6, 1994 May 6, 1995 June 30, 1995 October 10, 1995 October 10, 1995	February 28, 2007 October 9, 2007 July 27, 2009 March 10, 2008 July 15, 2010
H-share 0902 0995 1138 1071 0386 0953 0914 0177 0338 0177 0338 0177 0338 0177 0338 0177 0338 0177 0338 0177 0338 0177 0177 0177 0177 0177	1053 1088 0753 11288 1288
A-share 600011 600012 600012 600023 600023 600023 600023 600032 600032 600032 600332 600332 600332 600332 600332 600363 600363 600363 600363 600363 600363 600363 600363 600363 600363 600363 600363 600363 600377 600363 600363 600377 600363 600377 600376 6003777 6003777 60037777 60037777777777	601005 601088 601111 6011186 601288 601288
Company name Huaneng Power International Inc. Anhui Expressway Co. Ltd China Minsheng Banking Corp. Ltd China Minsheng Banking Corp. Ltd China Shipping Development Co. Ltd Huadian Power International Corp. Ltd China Petroleum & Chemical Corp. Ltd China Merchants Bank Co. Ltd China Merchants Bank Co. Ltd China Merchants Bank Co. Ltd China Bestern Airlines Corp. Ltd China Bestern Airlines Corp. Ltd China Bestern Airlines Corp. Ltd China Bestern Airlines Co. Ltd Gaangzhou Pharmaceutical Co. Ltd Jiangxi Copper Co. Ltd Jiangxi Copper Co. Ltd Jiangsu Expressway Co. Ltd Anhui Conch Cement Co. Ltd Anhui Conch Cement Co. Ltd Anhui Conch Cement Co. Ltd Shenzhen Expressway Co. Ltd Anhui Conch Cement Co. Ltd Shenzhen Expressway Co. Ltd Anhui Conch Cement Co. Ltd Shenzhan Bertronics Co. Ltd Anhui Conch Cement Co. Ltd Shenzhan Bertronics Co. Ltd Maanshan Iron & Steel Co. Ltd Sherig Panda Electronics Co. Ltd Sherig Panda Electronics Co. Ltd Sherig Panda Electronics Co. Ltd Sherig Panda Electronics Co. Ltd Maanshan Iron & Steel Co. Ltd Maanshan Iron & Steel Co. Ltd Sinoji Group Kumming Machinery Holdings Ltd Sinoyang Glass Co. Ltd Dongfang Electric Corp. Ltd Dongfang Electric Corp. Ltd Dongfang Electric Corp. Ltd	Chongqing from & Steel Co. Ltd China Shenhua Energy Co. Ltd Sichuan Expressway Co. Ltd Air China Ltd Air China Ltd Agricultural Bank of China Ltd Agricultural Bank of China Ltd Agricultural Bank of China Ltd
المنارك للاستشارات	www.

CFRI 3,1	Sector	Finance Finance Finance Finance Froperties Industrials Finance Conglomerates Finance Industrials Industrials Industrials Industrials Industrials Industrials Public utility Public utility Public utility Finance Finance Finance Finance Finance Finance Finance Finance
88	H-share IPO date	June 24, 2004 June 24, 2004 May 14, 1996 December 7, 2007 October 27, 2006 May 14, 1997 December 22, 2009 September 24, 2009 April 28, 2005 April 28, 2005 June 16, 2004 April 28, 2006 December 19, 2006 December 23, 2005 June 1, 2006 December 23, 2005 June 1, 2006 December 23, 2005 June 1, 2006 March 21, 1997 April 27, 2007
	A-share IPO date	March 1, 2007 May 15, 2007 December 22, 2006 December 27, 2006 October 16, 2006 April 30, 2007 September 25, 2007 January 9, 2007 December 5, 2008 September 5, 2007 December 5, 2007 December 12, 2007 December 5, 2007 December 12, 2007 July 5, 2006 July 5, 2006 April 27, 2007 December 25, 2007 December 25, 2007 July 5, 2006 April 27, 2007
	H-share	2318 2328 0525 0529 0529 0588 2500 1538 0588 2528 2528 1538 0539 1548 1548 1557 2885 2885 1568 1838 2856 2857 1568 1568 1568 1568 1568 1568 1568 1568
	A-share	601318 601328 601328 601339 601339 601588 601588 601588 601588 601588 601588 601588 601588 601588 601889 601889 601889 601889 601889 601889 601889 601889 601889 601889 601889 601899 6019399 6019398 6019399 6019398 6019398 6019398 6019399 6019398 6019399 6019398 6019399 6019398 6019399 6019398 6019398 6019398 6019399 6019398 6019398 6019399 6019398 6019398 6019398 6019399 6019398 6019398 6019399 6019398 6019398 6019399 6019398 6019398 6019398 6019398 6019399 6019399 6019399 6019398 60193999 6019399 6019999000000000000000000000000000000
Table AI	mpany name	ag An Insurance (Group) Co. of China Ltd ank of Communications Co. Ltd angshen Railway Co. Ltd ina Railway Group Ltd dustrial & Commercial Bank of China Ltd dustrial & Commercial Bank of China Ltd distrial & Commercial Bank of China Ltd ijing North Star Co. Ltd uminum Corp. of China Ltd argenific Insurance (Group) Co. of China Ltd anghai Electric Group Co. Ltd anghai Electric Group Co. Ltd anghai Electric Group Co. Ltd ina Diffeld Services Ltd troChina Co. Ltd ina Oilfield Services Ltd troChina Co. Ltd ina Colfing Container Lines Co. Ltd ina Shipping Container Lines Co. Ltd ina Colfings Co. Ltd ina Contruction Bank Corp. and Corp. Ltd ina Construction Bank Corp. It dina Drot (PDA) Co. Ltd ina Construction Bank Corp. ank of China Ltd ina CITIC Bank Corp. Ltd

المنسارات

	Append	lix	2
--	--------	-----	----------

المنارات

Company name	A-share code	H-share code
China Minsheng Banking Corp. Ltd	600016	1988
China Petroleum & Chemical Corp.	600028	0386
China Southern Airlines Co. Ltd	600029	1055
China Merchants Bank Co. Ltd	600036	3968
Jiangxi Copper Co. Ltd	600362	0358
China Shenhua Energy Co. Ltd	601088	1088
Air China Ltd	601111	0753
China Railway Construction Corp. Ltd	601186	1186
Ping An Insurance (Group) Co. of China Ltd	601318	2318
Bank of Communications Co. Ltd	601328	3328
China Railway Group Ltd	601390	0390
Industrial & Commercial Bank of China Ltd	601398	1398
Aluminum Corp. of China Ltd	601600	2600
China Pacific Insurance (Group) Co. of China Ltd	601601	2601
China Life Insurance Co. Ltd	601628	2628
Shanghai Electric Group Co. Ltd	601727	2727
CSR Corp. Ltd	601766	1766
PetroChina Co. Ltd	601857	0857
China Coal Energy Co. Ltd	601898	1898
Zijin Mining Group Co. Ltd	601899	2899
China COSCO Holdings Co. Ltd	601919	1919
China Construction Bank Corp.	601939	0939
Bank of China Ltd	601988	3988

Dual listed Chinese shares

89

 Table AII.

 The list of firms with

 shorted A-shares after the

 introduction of short-sale

 policy in A-share market

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

